COMMITTEE OF GOVERNORS OF THE CENTRAL BANKS OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC COMMUNITY

GROUP OF EXPERTS UNDER THE CHAIRMANSHIP OF MR. DALGAARD

# DRAFT TEXTS GIVING EFFECT TO THE AGREEMENT ON STRENGTHENING THE EMS

# 29тн МАҮ 1985

In accordance with the mandate given by the Committee of Governors at its meeting on 12th March 1985, the group of experts (see the attached list) has prepared the following draft texts giving effect to the Agreement on strengthening the EMS:

- Instrument relating to the Agreement of 13th March 1979 between the central banks of the Member States of the European Economic Community laying down the operating procedures for the European Monetary System;
- Decision (No. 17/85) amending Decision (No. 12/79) of the Board of Governors of the EMCF of 13th March 1979;
- Decision of the Board of Governors of the EMCF laying down the terms and conditions for acquisition, holding and use of ECUs by Other Holders;
- specimen of a letter from the Chairman of the Board of Governors of the EMCF to an Other Holder;
- entry in the minutes of the meetings of the Committee of Governors and of the Board of Governors of the EMCF on 10th June 1985;
- Sixth Directive from the Board of Governors to the Agent, replacing the Fourth and Fifth Directives.

In addition, in accordance with the mandate given by the Committee of Governors at its meeting on 14th May 1985, the group of experts has prepared the draft Instrument concerning the accession of the Bank of Greece to the Agreement of 13th March 1979, as amended by the Instrument dated 10th June 1985, between the central banks of the Member States of the European Economic Community laying down the operating procedures for the European Monetary System.

1

All texts other than the Decision laying down the terms and conditions for acquisition, holding and use of ECUs by Other Holders could be signed or adopted at the meeting of 10th June 1985. This Decision formally can be adopted only later when Regulation (EEC) No. 3181/78 has been amended by the Council of the European Communities in order to enable the EMCF to accord the status of Other Holder to non-EEC central banks and international monetary institutions. A provisional list of reference rates referred to in Article 8 of the EMS Agreement and an example illustrating the effects of the new formula on ECU remuneration is also attached for information.

Draft

### INSTRUMENT

RELATING TO THE AGREEMENT OF 13th MARCH 1979 BETWEEN THE CENTRAL BANKS OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC COMMUNITY LAYING DOWN THE OPERATING PROCEDURES FOR THE EUROPEAN MONETARY SYSTEM

THE CENTRAL BANKS OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC COMMUNITY -

Having regard to Article 21 of the Agreement of 13th March 1979 providing for a periodic review of the operation of the said Agreement in the light of experience gained;

Whereas it is desirable to strengthen the EMS by improving the conditions governing the use of the ECU issued by the EMCF;

Whereas it appears opportune to create the possibility of extending the circle of users of the ECU beyond the EEC central banks -

HAVE AGREED as follows:

### Article 1

The Agreement of 13th March 1979 between the Central Banks of the Member States of the European Economic Community laying down the operating procedures for the European Monetary System (hereinafter referred to as the "Agreement of 13th March 1979") shall be amended as follows:

(a) The text of Article 8 shall be replaced by the following text:

"8.1 The debtor and creditor interest rates applying to very short-term financing operations shall be the weighted average of the most representative domestic money-market rates in those EEC countries whose currencies make up the ECU basket. The average is weighted in accordance with the weights of these currencies in the ECU basket as derived from the ruling ECU central rates. The weighted average shall be calculated on the basis of the average of daily observations during each month and shall apply during the following month to all outstanding amounts in respect of very short-term financing operations.

8.2 Accrued interest shall be paid in ECUs at each monthly settlement date or, between settlement dates, at the same time as advance liquidation of a debtor balance is effected.

8.3 For the purposes of Article 8.1 each central bank whose currency is in the ECU basket shall notify the Secretariat of the Committee of Governors of the Central Banks of the Member States of the European Economic Community of the title of its reference rate. The Secretariat shall pass on this information to the other central banks, the Commission of the European Communities and the Agent of the European Monetary Co-operation Fund."

(b) The text of Article 16.1, first sub-paragraph, shall be replaced by the following text:

"When a financing operation falls due, settlement shall be carried out - insofar as this is not done in the first instance by means of holdings in the creditor's currency - entirely or in part by transferring ECUs, with the proviso that:

- a creditor central bank shall not be obliged to accept settlement by means of ECUs of an amount equal to more than 50 per cent. of its claim which is being settled, unless its assets in ECUs are smaller than its forward sales of ECUs to the EMCF;
- if, but only to the extent that, a creditor central bank's assets in ECUs are smaller than its forward sales of ECUs to the EMCF, it shall be obliged to accept full settlement in ECUs; if the claim of the central bank in question exceeds the amount of its net debtor position in ECUs, the rule set out in the above indent shall be applied to the balance of the claim.

- 2 -

ECU assets referred to above include forward purchases of ECUs from the EMCF.

Insofar as settlement is only partially effected by transferring ECUs, the balance shall be settled by transferring other reserve assets in accordance with the composition of the debtor central bank's reserves as at the end of the month preceding the settlement."

(c)

The text of Article 18.2 shall be replaced by the following text:

"18.2 The EEC central banks may transfer ECUs to one another against currencies, gold and other monetary assets. They may also exchange ECUs with other central banks and international monetary institutions which have been accorded the status of Other Holder by the European Monetary Co-operation Fund on the terms and conditions it has established."

(d)

A new Article 18a shall be inserted, worded as follows:

### "18a - Mobilisation mechanism

18a.1 In the event of a need for intervention currency a central bank may mobilise ECU holdings against dollars through the intermediary of the EMCF up to a ceiling which is defined in the following way:

- a proportion of the ECUs received by the central bank concerned through the operations referred to in Article 17.1 of the present Agreement. This proportion shall be determined at the beginning of each swap period by the ratio of the ECU equivalent of the EMCF's total dollar holdings to the total amount of ECUs created by the EMCF under Article 17.1 of the present Agreement, multiplied by a factor which shall be agreed unanimously by the Governors of the central banks,
- plus the amount of ECU assets held by the central bank in excess of its forward sales of ECUs to the EMCF (net creditor position), or minus the amount by which its forward sales of ECUs to the EMCF exceed its ECU assets (net debtor position). ECU assets referred to above include forward purchases of ECUs from the EMCF.

At least four working days' notice shall be given by the mobilising central bank.

- 3 -

18a.2 To cover the mobilisation operations referred to in the preceding paragraph, each central bank other than the mobilising central bank undertakes to provide spot dollars in proportion to and up to an amount equivalent to its forward purchases of dollars from the EMCF in accordance with Article 17 of the present Agreement, minus any amount which may have been withdrawn in accordance with Article 18.3. In exceptional circumstances, a central bank may opt out entirely or in part from participation in the initial mobilisation operation (see Article 18a.1) or its renewal (see Article 18a.5). In such an event the uncovered amount shall be shared out in proportion to the undertakings of the other central banks. To the extent that a mobilisation request cannot be met out of the commitments of the other central banks, it shall be automatically scaled down by the necessary amount.

18a.3 At the request of the mobilising central bank and through the intermediary of the EMCF, the proceeds of the mobilisation operations referred to above may be converted immediately fully or in part into the currency of a participating central bank, subject to its agreement. In this event, the conversion request will be met by the central bank issuing the currency requested, together with any other central bank making voluntary contributions.

18a.4 All the operations with the EMCF referred to in the preceding paragraphs shall take the form of three-month swaps which shall be concluded at flat rates. The rate applicable is the market rate two working days prior to the value date of the spot transactions.

18a.5 In the event of continuing need for intervention currency the swaps referred to above may be renewed once at the request of the mobilising central bank. In this event, the EMCF and the central banks shall make the necessary adjustments

- first, to ensure that a central bank's use of the mobilisation mechanism does not exceed the ceiling referred to in Article 18a.1,
- secondly, to ensure that the dollar contributions made by the central banks in respect of Article 18a.2 are in proportion to and within the commitments defined there,

- 4 -

- thirdly, to take account of exchange rate changes that may have occurred since the beginning of the swap to be renewed.

18a.6 At the request of the mobilising central bank the swap may be unwound fully or in part before maturity with at least three working days' notice.

18a.7 Central banks receiving currency assets from the EMCF shall pay interest on these assets. Central banks which have provided currency assets to the EMCF shall receive interest on these assets. The rate of interest shall be the most representative domestic money-market rate in the country whose currency has been used two working days prior to the value date of the spot transaction, i.e. in the case of the US dollar the rate used by the IMF for the calculation of SDR remuneration and in the case of Community currencies the rates provided for by Article 8.1 of the present Agreement. Interest payments shall be due at maturity of the swaps or when the swaps are unwound prematurely and shall be payable in the currency used."

(e) Article 19 shall be supplemented as follows:

"19.3 ECU assets referred to in Article 19.1 include forward purchases of ECUs from the EMCF."

"19.4 Articles 19.1 and 19.2 apply by analogy to remuneration of ECU assets held with the EMCF by Other Holders referred to in Article 18.2 of the present Agreement."

(f) The text of Article 20.1 shall be replaced by the following text:

"20.1 Save in the event of a unanimous decision to the contrary, the swaps referred to in Articles 17.3 and 18a.4 of the present Agreement shall be unwound at the end of the period for which the Governors of the central banks extend the mechanism provided for in Article 17 of the present Agreement."

#### Article 2

The present instrument shall enter into force as from 1st July 1985, with the proviso that the amendment to Article 18.2 of the Agreement

- 5 -

of 13th March 1979 shall not take effect until the date on which the European Monetary Co-operation Fund is enabled to accord the status of Other Holder to other central banks and to international monetary institutions.

### Article 3

The present instrument shall be drawn up and duly signed in English, French and German. One certified copy of the original in each language shall be sent to each central bank by the Secretariat of the Committee of Governors, which is required to retain the originals.

Done at Basle, 10th June 1985

Banque Nationale de Belgique

Danmarks Nationalbank

Deutsche Bundesbank

Banque de France

Central Bank of Ireland

Banca d'Italia

.

.

De Nederlandsche Bank

Bank of England

# INSTRUMENT RELATING TO THE ACCESSION OF THE BANK OF GREECE

### TO THE

#### AGREEMENT

BETWEEN THE CENTRAL BANKS OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC COMMUNITY LAYING DOWN THE OPERATING PROCEDURES FOR THE EUROPEAN MONETARY SYSTEM

THE GOVERNORS OF THE CENTRAL BANKS OF THE MEMBER STATES OF THE EUROPEAN ECONOMIC COMMUNITY, having met this day in Basle -

Whereas on 1st January 1981 Greece became -a member of the European Communities and party to the Treaties setting up these Communities;

Whereas the Greek drachma has been included in the ECU basket since 17th September 1984;

Whereas the Bank of Greece has signified its desire to accede to the Agreement of 13th March 1979, as amended by the Instrument dated 10th June 1985, between the central banks of the Member States of the European Economic Community laying down the operating procedures for the European Monetary System (called hereafter the "Agreement"), without, however, participating initially in the exchange rate mechanism provided for in the Agreement;

Whereas Article 17 of the Agreement provides for voluntary contributions of gold and dollars against ECUs to the European Monetary Cooperation Fund by central banks which do not participate in the exchange rate mechanism and whereas the Bank of Greece has declared its intention to make use of this possibility in the near future -

HAVE AGREED on the following provisions:

I. The Bank of Greece becomes party to the Agreement.

II. The text of the Agreement is included with the present Instrument and forms an integral part of it.

III. The present Instrument shall enter into force on 1st July 1985, one copy having been drawn up and duly signed in English, French and German and the text referred to in Section II above having been duly initialled. One certified copy of the original in each language shall be sent to each central bank by the Secretariat of the Committee of Governors, which is required to retain the originals.

Done at Basle, 10th June 1985

Banque Nationale de Belgique

Danmarks Nationalbank

Deutsche Bundesbank ·

Τράπεζα της Ελλάδος

Banque de France

Central Bank of Ireland

Banca d'Italia

- --

De Nederlandsche Bank

Bank of England

.

### EUROPEAN MONETARY CO-OPERATION FUND

Draft

# DECISION (No. 17/85) AMENDING DECISION (No. 12/79) OF THE BOARD OF GOVERNORS OF 13th MARCH 1979

THE BOARD OF GOVERNORS OF THE EUROPEAN MONETARY CO-OPERATION

Having regard to Regulation (EEC) No. 907/73 of the Council of the European Communities of 3rd April 1973 establishing a European Monetary Co-operation Fund, and in particular Articles 3 and 4 therof;

Having regard to Regulation (EEC) No. 3181/78 of the Council of the European Communities of 18th December 1978 concerning the European Monetary System, and in particular Article 3 thereof;

Having regard to the amendments made by the EEC central banks to their Agreement of 13th March 1979;

Pending adoption of the Regulation amending Regulation (EEC) No. 3181/78 of the Council of the European Communities enabling the EMCF to accord to non-EEC central banks and international monetary institutions the status of Other Holder of ECUs issued by the EMCF -

HAS DECIDED as follows:

ARTICLE 1

The Decision (No. 12/79) shall be amended as follows:

(a) In Article 1, the words "as amended by the Instrument of 10th June May 1985" will be inserted after "Agreement of 13th March 1979".

(b) The text of Article 3.2 shall be replaced by the following text:

"3.2 The EEC central banks may transfer ECUs to one another against currencies, gold and other monetary assets. They may also exchange

ECUs with other central banks and international monetary institutions which have been accorded the status of Other Holder by the European Monetary Co-operation Fund on the terms and conditions it has established."

# (c) A new Article 3a shall be inserted, worded as follows: "3a - Mobilisation mechanism

3a.1 In the event of a need for intervention currency a central bank may mobilise ECU holdings against dollars through the intermediary of the EMCF up to a ceiling which is defined in the following way:

- a proportion of the ECUs received by the central bank concerned through the operations referred to in Article 2.1 of the present Decision. This proportion shall be determined at the beginning of each swap period by the ratio of the ECU equivalent of the EMCF's total dollar holdings to the total amount of ECUs created by the EMCF under Article 2.1 of the present Decision, multiplied by a factor which shall be agreed unanimously by the Governors of the central banks,
- plus the amount of ECU assets held by the central bank in excess of its forward sales of ECUs to the EMCF (net creditor position), or minus the amount by which its forward sales of ECUs to the EMCF exceed its ECU assets (net debtor position). ECU assets referred to above include forward purchases of ECUs from the EMCF.

At least four working days' notice shall be given by the mobilising central bank.

3a.2 To cover the mobilisation operations referred to in the preceding paragraph, each central bank other than the mobilising central bank undertakes to provide spot dollars in proportion to and up to an amount equivalent to its forward purchases of dollars from the EMCF in accordance with Article 2 of the present Decision, minus any amount which may have been withdrawn in accordance with Article 3.3. In exceptional circumstances, a central bank may opt out entirely or in part from participation in the initial mobilisation operation (see Article 3a.1) or its renewal (see Article 3a.5). In such an event the uncovered amount shall be shared out in proportion to the undertakings of the other central banks. To the extent that a mobilisation request cannot be met out of the commitments of the other central banks, it shall be automatically scaled down by the necessary amount.

- 2 -

3a.3 At the request of the mobilising central bank and through the intermediary of the EMCF, the proceeds of the mobilisation operations referred to above may be converted immediately fully or in part into the currency of a participating central bank, subject to its agreement. In this event, the conversion request will be met by the central bank issuing the currency requested, together with any other central bank making voluntary contributions.

3a.4 All the operations with the EMCF referred to in the preceding paragraphs shall take the form of three-month swaps which shall be concluded at flat rates. The rate applicable is the market rate two working days prior to the value date of the spot transactions.

3a.5 In the event of continuing need for intervention currency the swaps referred to above may be renewed once at the request of the mobilising central bank. In this event, the EMCF and the central banks shall make the necessary adjustments

- first, to ensure that a central bank's use of the mobilisation mechanism does not exceed the ceiling referred to in Article 3a.1;
- secondly, to ensure that the dollar contributions made by the central banks in respect of Article 3a 2. are in proportion to and within the commitments defined there;
- thirdly, to take account of exchange rate changes that may have occurred since the beginning of the swap to be renewed.

3a.6 At the request of the mobilising central bank the swap may be unwound fully or in part before maturity with at least three working days' notice.

3a.7 Central banks receiving currency assets from the EMCF shall pay interest on these assets. Central banks which have provided currency assets to the EMCF shall receive interest on these assets. The rate of interest shall be the most representative domestic money-market rate in the country whose currency has been used two working days prior to the value date of the spot transaction, i.e. in the case of the US dollar the rate used by the IMF for the calculation of SDR remuneration and in the case of Community currencies the rates provided for by Article 8.1 of the

- 3 -

Agreement of 13th March 1979. Interest payments shall be due at maturity of the swaps or when the swaps are unwound prematurely and shall be payable in the currency used."

(d) The text of Article 4.1, first sub-paragraph, shall be replaced by the following text:

"When a financing operation falls due, settlement shall be carried out - insofar as this is not done in the first instance by means of holdings in the creditor's currency - entirely or in part by transferring ECUs, with the proviso that:

- a creditor central bank shall not be obliged to accept settlement by means of ECUs of an amount equal to more than 50 per cent. of its claim which is being settled, unless its assets in ECUs are smaller than its forward sales of ECUs to the EMCF;
- if, but only to the extent that, a creditor central bank's assets in ECUs are smaller than its forward sales of ECUs to the EMCF, it shall be obliged to accept full settlement in ECUs; if the claim of the central bank in question exceeds the amount of its net debtor position in ECUs, the rule set out in the above indent shall be applied to the balance of the claim.

ECU assets referred to above include forward purchases of ECUs from the EMCF.

Insofar as settlement is only partially effected by transferring ECUs, the balance shall be settled by transferring other reserve assets in accordance with the composition of the debtor central bank's reserves as at the end of the month preceding the settlement."

(e) Article 5 shall be supplemented as follows:

"5.3 ECU assets referred to in Article 5.1 include forward purchases of ECUs from the EMCF."

"5.4 Articles 5.1 and 5.2 apply by analogy to remuneration of ECU assets held with the EMCF by Other Holders referred to in Article 3.2 of the present Decision."

(f) The text of Article 6.1 shall be replaced by the following text:

- 4 -

"6.1 Save in the event of a unanimous decision to the contrary, the swaps referred to in Articles 2.3 and 3a.4 above shall be unwound at the end of the period for which the Governors of the central banks extend the mechanism provided for in Article 2 of the present Decision."

### ARTICLE 2

The present Decision shall enter into force as from 1st July 1985, with the proviso that the amendment to Article 3.2 of the Decision (No 12/79) shall not take effect until the date on which the European Monetary Co-operation Fund is enabled to accord to other central banks and to international monetary institutions the status of Other Holder of ECUs.

Done at Basle, 10th June 1985

On behalf of the Board of Governors

The Chairman

#### EUROPEAN MONETARY CO-OPERATION FUND

DECISION (No. ../85) OF THE BOARD OF GOVERNORS LAYING DOWN THE TERMS AND CONDITIONS FOR ACQUISITION, HOLDING AND USE OF ECUS BY OTHER HOLDERS Draft

THE BOARD OF GOVERNORS OF THE EUROPEAN MONETARY CO-OPERATION

Having regard to Regulation (EEC) No. 907/73 of the Council of the European Communities of 3rd April 1973 establishing a European Monetary Co-operation Fund, and the statutes annexed thereto,

Having regard to Article ... of Regulation (EEC) No. 3181/78 of the Council of the European Communities of 18th December 1978 relating to the European Monetary System, as amended by Regulation (EEC) No. ... of the Council of the European Communities of ..... 1985,

Whereas it is necessary to determine the terms and conditions on which non-EEC central banks and international monetary institutions which have been accorded the status of Other Holder by the Board of Governors of the European Monetary Co-operation Fund (hereafter called EMCF), may acquire, hold and use ECUs issued by the EMCF -

HAS DECIDED as follows:

Article 1 Designation of Other Holder

The Board of Governors of the EMCF may accord the status of Other Holder to non-EEC central banks and international monetary institutions at their request.

Article 2 Transactions between Other Holders and EEC central banks 2.1 An Other Holder may acquire ECUs on a voluntary basis against other monetary assets from an EEC central bank or an Other Holder either in a swap transaction or in a transaction under a repurchase agreement. It may use its ECU holdings either (a) to unwind these operations or (b) in swap transactions or in transactions under a repurchase agreement with EEC central banks or Other Holders. Transactions referred to under (b) are subject to the provision that they shall mature at a date not later than the maturity date of the transactions through which the Other Holder effecting spot sales has obtained the ECUs concerned.

All operations referred to above shall be concluded subject to the condition that, at the latest at the end of the month in which the operation is terminated, interest accrued in ECUs must be sold back to the institution (EEC central bank or Other Holder) which has effected spot sales of ECUs in connection with the swap or the transaction under the repurchase agreement. Interest accrued is defined as (a) interest received by the Other Holder in accordance with Article 3.1 below, and (b) interest which has been sold back to the Other Holder in application of the provision referred to in the preceding sentence.

2.2 An EEC central bank may conclude transactions with Other Holders as specified under 2.1 above.

2.3 All other terms and conditions of these transactions will be mutually agreed by the parties involved.

Article 3 Holdings of ECUs by Other Holders

3.1 ECUs acquired in accordance with Article 2 and received as interest paid on outstanding balances will be held by Other Holders on the same terms and conditions prevailing at that time for EEC central banks and established by the Board of Governors of the EMCF.

3.2 Other Holders are not required to reimburse expenses incurred by the EMCF in administering the ECU accounts.

# Article 4 Mutual information

4.1 The EMCF will inform Other Holders of all matters relevant to their transactions in and holdings of ECUs.

- 2 -

4.2 An Other Holder shall inform the Agent of the EMCF of transactions concluded with EEC central banks and Other Holders in compliance with the relevant provisions of the Annex to this Decision which regulates the relations between the Agent and the Other Holders.

# Article 5 Termination of the status of Other Holder

5.1 The status of Other Holder of ECUs may be terminated by a decision of the Board of Governors of the EMCF or by written notice from the Other Holder to the Secretariat of the Board of Governors of the EMCF. In this event all outstanding operations referred to in Article 2 of this Decision will have to be unwound as early as possible.

5.2 Independently from the provisions of the preceding paragraph, the status of Other Holder will be terminated in the event of liquidation of the mechanism for creation of ECUs provided for by Article 6 of the Decision (No. 12/79) of the Board of Governors of the EMCF of 13th March 1979, as amended by Decision (No. 17/85) of 10th June 1985. In this case all outstanding operations referred to in Article 2 of this Decision will have to be unwound immediately, irrespective of their agreed maturity.

### Article 6 Review and revision

The Board of Governors of the EMCF may review this Decision and, if necessary, revise it.

### Article 7 Application

The present Decision and any amendments to it will form an integral part of any agreements between the EMCF and Other Holders. Any changes to this Decision will apply to all operations concluded thereafter between EEC central banks and Other Holders.

Done at Basle, ..... 1985

On behalf of the Board of Governors The Chairman

- 3 -

### ANNEX

to Decision (No. ../85) of the Board of Governors of the European Monetary Co-operation Fund laying down the terms and conditions for acquisition, holding and use of ECUs by Other Holders

#### RULES

governing relations between the Agent of the EMCF ("the Agent") and Other Holders of ECUs.

Article 1

1.1 All communications between the Agent and the Other Holders shall be in English, French or German; they shall be transmitted by duly numbered and authenticated telex, telefax or telegram.

1.2 A special system of numbering shall be reserved for these messages; the serial number of each of them shall be immediately preceded by the code word FECOM, to ensure that on arrival it is delivered solely to those persons responsible for transactions with the Fund.

### Article 2

An Other Holder shall inform the Agent of all transactions concluded with EEC central banks and Other Holders; such information shall cover the amount and the value date of the transactions.

### Article 3

ς.

3.1 At the beginning of each month the Agent will notify the Other Holders of the interest rate applicable to their ECU balances. This interest rate will be calculated by the Agent in accordance with Article 8 of the Agreement between EEC central banks laying down the operating procedures for the European Monetary System, as amended on 10th June 1985.

3.2 Interest on the balances on Other Holders' accounts will be calculated as at the last working day of each month or as at the time when an account is liquidated, on the basis of the monthly rate referred to above, the exact number of days and 365 days a year.

3.3 Interest thus calculated will be settled in ECUs by crediting the account of the Other Holder concerned on the last working day of each month.

### Article 4

The Agent will give the Other Holder concerned confirmation of the execution of every accounting operation, indicating at the same time the account balance, and it will send them by post a monthly statement of account.

Draft

# SPECIMEN OF A LETTER FROM THE CHAIRMAN OF THE BOARD OF GOVERNORS OF THE EMCF TO AN OTHER HOLDER

Following your request dated . . . . , the Board of Governors of the EMCF at its meeting on . . . . decided to accord to the (name of Other Hölder) the status of Other Holder of ECUs. The terms and conditions for acquisition, holding and use of ECUs by Other Holders are laid down in the attached Decision. The Board will inform the (name of Other Holder) of any amendments which may be made to this Decision; in the event of such amendments being made the (name of Other Holder) will be asked to indicate whether or not it wishes to retain the status of Other Holder.

I should be grateful if you would write to confirm your acceptance of the conditions set out in this Decision and propose that in that event this letter and your reply to it should constitute an agreement between our two Institutions. DRAFT ENTRY IN THE MINUTES OF THE MEETING OF THE COMMITTEE OF GOVERNORS ON 10th JUNE 1985

.

ν.:

The Governors decide unanimously to fix the multiplier referred to in Article 18a.1 of the Agreement of 13th March 1979, as amended by the instrument dated 10th June 1985, at 1.5. The multiplier will be re-examined at the request of any participating central bank.

# DRAFT ENTRY IN THE MINUTES OF THE MEETING OF THE BOARD OF GOVERNORS OF THE EMCF ON 10th JUNE 1985

In conformity with the decision of the Governors of the EEC central banks at their meeting on 10th June 1985, the Board of Governors fixes the multiplier referred to in Article 3a.1 of the Decision (No. 12/79) of 13th March 1979, as amended by Decision (No. 17/85) of 10th June 1985, at 1.5. The multiplier will be re-examined at the request of any participating central bank.

### EUROPEAN MONETARY CO-OPERATION FUND

Draft

### SIXTH DIRECTIVE FROM THE BOARD OF GOVERNORS TO THE AGENT, DATED 10th JUNE 1985

This Directive, based on Decision No. 1/73 of the Board of Governors of the European Monetary Co-operation Fund (hereinafter referred to as the "Board" and the "Fund" or the "EMCF") of 14th May 1973, as amended on 13th March 1979 and 10th June 1985,

- governs, in the context of the Fund's operations, the relations between the Agent, on the one hand, and the Board, the central banks represented on the Board (hereinafter referred to as the "central banks") and Other holders of ECUs, on the other; and
- sets out the tasks entrusted to the Agent by the EMCF pursuant to the Agreement of 13th March 1979 between the Central Banks of the Member States of the European Economic Community (hereinafter referred to as the "EEC") laying down the operating procedures for the European Monetary System (hereinafter referred to as the "Agreement" and the "EMS") and in accordance with
  - . Decision No. 12/79 of the Board, dated 13th March 1979, as amended by Decision No. 17/85 of 10th June 1985, and
  - . the Agreement concerning short-term monetary support, dated 9th February 1970.

With the entry into force of the present Directive on 1st July 1985, the Fourth and Fifth Directives to the Agent dated 13th March 1979 shall lapse.

#### I. GENERAL PROVISIONS

### Article 1 - Functions of the Agent

# 1.1 Conclusion of swap operations in ECUs

(see Sections III of the Agreement and of the present Directive) In the name and for the account of the Fund, the Agent shall conclude with the central banks the swap operations referred to in Articles 2 and 3a of Decision No. 12/79.

#### 1.2 Management contract

Whenever any swap operation as referred to in Article 17.3 of the Agreement is concluded or renewed, the Agent shall, in the name of the Fund, entrust the central banks concerned with the management of the reserve assets transferred to the Fund.

### 1.3 Accounting procedures in the books of the EMCF

The Agent shall keep the Fund's accounts, in ECUs as defined by Article 1 of Council Regulation (EEC) No. 3180/78 of 18th December 1978 as amended from time to time, intended for recording:

- in connection with the very short-term intra-Community facilities and under the short-term monetary support arrangements, the central banks' debts and claims vis-à-vis the Fund arising from operations in member countries' currencies on their foreign exchange markets,
- in connection with the creation of ECUs, the claims and liabilities arising from the swap operations,
- the positions arising from ECU mobilisation operations,
- the positions arising from operations between the central banks and Other holders of ECUs,
- the periodic settlements of principal and payments of interest in respect of these various debts and claims,
- transfers of ECUs between the central banks, on the one hand, and between the central banks and Other holders of ECUs, on the other, and
- the payment of interest on net ECU positions.

In the case of transactions effected

- in member countries' currencies under certain swap operations in connection with the mobilisation of ECUs, and
- in currencies of non-EEC countries under the short-term monetary support arrangements and swap operations in connection with the mobilisation of ECUs,

the accounts of the Fund shall be kept in the currencies actually involved in the borrowing or swap operations.

- 2 -

### Article 2 - Responsibility of the Agent

### 2.1 Limitation of the Agent's responsibility

The Agent shall be responsible only for the proper and faithful performance of the functions described in Articles 1, 3.3, 4, 14, 17.5 and 25 of the present Directive and of the instructions it receives for this purpose from the Board or the Chairman of the Board.

#### 2.2 Further instructions

The Agent may at any time, in particular if there is any doubt as to the extent of its obligations, ask for further instructions from the Board or, if it is a matter of urgency, from the Chairman of the Board.

#### 2.3 Discharge of the Agent

Receipt by the Board of any report and any statement of account submitted by the Agent shall discharge the latter from liability in respect of the operations described in the respective documents; this discharge shall be automatic if no reservations are expressed by the Board or by one of its members at the next Board meeting following that during which the documents in question were presented.

### 2.4 Further instructions from the Board

After receiving the documents referred to in Article 2.3 above the Board or, if it is a matter of urgency, the Chairman of the Board shall be responsible for giving the Agent any further instructions that may be required in that connection.

### Article 3 - Communications

# 3.1 Communications by cable, telex or telefax

All communications between the Agent and the central banks or Other holders of ECUs shall be in English, French or German and shall be transmitted by duly numbered and authenticated cable, telex or telefax.

#### 3.2 FECOM numbering

A special system of numbering shall be reserved for these messages, the serial number of each of them being immediately preceded by the word "FECOM", to ensure that they are delivered solely to those persons responsible for transactions with the Fund.

### 3.3 Accounting communications

The Agent shall

- send the central banks or Other holders of ECUs concerned confirmation of the execution of every accounting operation, indicating at the same time the balance on the accounts affected by the operations in question, and
- at the beginning of each month send them by mail a statement of their accounts as at the last working day of the previous month.

### Article 4 - Reports by the Agent

# 4.1 Administrative report and statement of account

In the first week of each month the Agent shall submit to the Board a report on the operations recorded by the Fund in the course of the preceding month, together with a statement of the accounts as at the end of that month.

# 4.2 Half-yearly summary report on the Fund's operations

With a view to the preparation of the half-yearly reports submitted by the Board of the Fund to the Council and to the Commission of the European Communities (hereinafter referred to as the "CEC"), the Agent shall provide the Secretariat of the Board, before 15th January and 15th July each year, with a summary of the operations recorded by the Fund during the preceding six months.

### 4.3 Supplementary report

The Agent shall have the right to send a supplementary report to the Board or to its Chairman any time it deems this necessary for the proper performance of its functions as set out in the present Directive.

### Article 5 - Exchange rates

### 5.1 Central rates of the ECU

The Secretariat of the Committee of Governors of the Central Banks of the Member States of the EEC shall notify the Agent of the central rates of the ECU vis-à-vis the currencies of the EEC countries.

### 5.2 Daily rates for the ECU

The Agent shall receive from the CEC the daily rate for the ECU in each of the EEC currencies and also in other currencies.

- 4 -

### Article 6 - Remuneration

(see Article 8 (amended) and Article 19 of the Agreement)

### 6.1 Representative interest rates

Each central bank shall notify the Agent on the last working day of the month of the monthly average of its most representative domestic moneymarket interest rate, calculated to two decimal places.

6.2 <u>Calculation and notification of the interest rate by the Agent</u> On the first working day of each month, the Agent shall

- on the basis of the interest rates reported by the central banks, calculate and
- inform the central banks and Other holders of ECUs of

the interest rate applying for the whole of that month to accounts representing very short-term financing operations, to the net positions in ECUs and to the "ECU reserves" accounts of Other holders of ECUs.

### 6.3 Formula for the interest rate calculation

The rate of interest applying to the accounts mentioned in Article 6.2 above shall be equal to the arithmetic mean of the rates referred to in the same Article, weighted in accordance with the weights of the respective currencies in the basket making up the ECU and rounded down to the nearest 1/4 per cent.; these weights shall be calculated on the basis of the central rates of the ECU ruling at the end of the previous month.

6.4 Interest rates applying to ECU mobilisation operations

On the occasion of ECU mobilisation and conversion operations in EMS currencies (see Section IV below), the Agent shall

- ascertain the US dollar interest rate used by the International Monetary Fund to calculate the rate of remuneration of the SDR;
- request the central bank or central banks concerned to notify the Agent of the interest rate or rates referred to in Article 6.1 above and ruling two days prior to the value date of the swap transaction.

6.5 Interest rate applying to short-term monetary support

The interest rate applying to accounts representing short-term monetary support operations shall be reported to the Agent by the Secretariat of the Committee of Governors.

### II. <u>VERY SHORT-TERM INTRA-COMMUNITY FACILITIES</u> (see Section II of the Agreement)

Article 7 - Very short-term financing operations

#### 7.1 Notification by the central banks

Very short-term financing operations arising from interventions in EEC currencies shall be notified by the central bank carrying them out:

- to the Agent, and

- to the central bank whose currency is involved.

Such notification shall be given in accordance with Article 3 above.

#### 7.2 Form of the notification

Such notification shall cover:

- the nature of the intervention (buying or selling),
- the amount expressed in the currency of the partner,
- the ECU countervalue of this amount, calculated at the ECU conversion rate referred to in Article 5.2 of the present Directive ruling on the day on which the operation is concluded,
- the spot operation value date, and
- the initial settlement date of the very short-term financing (last working day preceding the sixteenth day of the second month following the month in which the value date of the intervention falls, hereinafter referred to as "monthly settlement date" see Article 9.2 below).

7.3 Confirmation of the operation

The central bank issuing the currency involved in the intervention shall acknowledge receipt of the notification to the central bank that has intervened and shall confirm its agreement to the Agent in the same form as the initial notification.

### Article 8 - Accounts in respect of very short-term financing operations (see Article 7 of the Agreement)

The Agent shall open in the books of the Fund the following categories of accounts denominated in ECUs:

(a) initial debtor accounts, to record debts arising out of financing operations with an initial settlement date in accordance with Article 9 of the Agreement; these accounts shall be kept separately according to initial settlement date of the financing;

- (b) renewed debtor accounts, to record debts that have been renewed automatically in accordance with the provisions of Article 10 of the Agreement; these accounts shall be kept separately according to extended settlement date of the financing;
- (c) creditor accounts, to record the claims forming the counterpart of the debts entered in the categories of accounts mentioned under (a) and (b) of the present Article; these accounts shall be kept separately according to the month in which the value date of the intervention falls; no central bank may simultaneously hold claims on a creditor account and debts on an initial or renewed debtor account;
- (d) "mutual agreement" accounts, to record both debit and credit entries in respect of financing operations that have been extended in accordance with Article 11 of the Agreement; these accounts shall be kept separately according to the contractual settlement date of the financing.

#### Article 9 - Remuneration

(see Article 8 of the Agreement)

### 9.1 Basis for calculating interest

Interest on the overall balances on each of the categories of ECUdenominated accounts referred to in Article 8 above shall be calculated, as appropriate, as at the last working day of each month, the monthly settlement date or the date on which an account is liquidated in advance, on the basis of the rate referred to in Article 6.3 of the present Directive, the exact number of days and 365 days a year.

#### 9.2 Monthly settlement date

"Monthly settlement date" shall be taken to mean the actual maturity date of a debtor account (initial account, renewed account, "mutual agreement" account); the settlement date in respect of interest payable by the debtor shall determine the date on which the creditor shall collect it. 9.3 Terms and conditions for the settlement of interest

Upon repayment of a debt at the monthly settlement date or upon advance liquidation of a debtor account, the interest shall be added to the amounts of principal repaid and shall be settled in the same way.

### 9.4 Settlement of interest in ECUs at the monthly settlement date

If a debtor central bank does not make any repayment of principal at a monthly settlement date, the interest payable by it at that date shall be settled in ECUs by means of a transfer carried out by the Agent from the "ECU reserves" account of the debtor to that of the creditor.

### Article 10 - Accounting procedures (see Article 12 of the Agreement)

### 10.1 Transfers of currencies

Upon conclusion of a new very short-term financing operation or upon advance repayment by transfer in a creditor's currency, the creditor and debtor central banks shall, on the agreed value date, themselves execute the transfers of the currencies in question direct.

### 10.2 ECU accounting procedures

The Agent shall book the operations referred to in Article 10.1 above on their spot value date in ECUs using the conversion rate for the day on which the operation was concluded, as referred to in Article 5.2 of the present Directive, and according to the following procedures:

- by debiting the creditor accounts, in decreasing order of seniority and, if the balance on these accounts is nil, by debiting initial debtor accounts,
- by crediting the renewed debtor accounts, in the order of proximity of their settlement dates; if the balance on these accounts is nil, by crediting the initial debtor accounts, in the order of proximity of their settlement dates; and if the balance on the latter accounts is nil, by crediting creditor accounts.

### 10.3 Derogation

With the consent of all the creditor and debtor parties, the Agent may, notwithstanding the provisions of Article 10.2 above, debit or credit a "mutual agreement" account by way of voluntary offsetting, making the appropriate transfer where necessary.

Article 11 - Automatic renewal and renewal by "mutual agreement" (see Articles 10 and 11 of the Agreement)

### 11.1 Notification of renewal

A debtor central bank shall notify the Agent at least five working days before the initial settlement date of its intention to make use of the automatic renewal facility provided for in Article 10 of the Agreement; similarly, any agreement concluded with a view to renewal by mutual agreement in accordance with Article 11 of the said Agreement shall be notified to the Agent at least five working days before the settlement date on which it is to take effect.

### 11.2 Conformity with the Agreement

The Agent shall verify that the renewal requested or agreed upon is in conformity with the provisions of the Agreement; in case of doubt, it shall inform the Chairman of the Board.

### 11.3 Transfers in the books of the Fund

The Agent shall make the necessary transfers in the books of the Fund to give effect to the renewal on the relevant monthly settlement date.

# Article 12 - Confirmation of accounting operations

The Agent shall give confirmation of the execution of the accounting operations to the central banks concerned and inform them of the overall balance on each category of accounts, as well as giving a breakdown according to accounting periods.

# Article 13 - Reserves of the central banks

# 13.1 Notification of the Agent by the central banks

13.1.1 Before the 5th of each month, the central banks shall send the Agent a statement as at the end of the previous month of their gross assets, excluding contributions of US dollars to the EMCF, in

- IMF reserve positions,
- special drawing rights,
- US dollars,
- other currencies,

expressed in ECUs and in US dollars; conversion of these assets into ECUs and US dollars shall be effected at the rates or values obtaining on the third last working day of the month to which the statement refers. 13.1.2 This statement shall also include mention of the assets held by the central banks in gold, expressed in thousands of fine ounces and in fine kilogrammes, likewise excluding contributions to the EMCF.

### 13.2 Summary statement sent by the Agent

As soon as it is in possession of the notifications in question, the Agent shall send each central bank, before the 10th of each month, a summary statement of these reserves which shall also mention the balances on "ECU reserves" accounts and any positions in the books of the EMCF in respect of very short-term financing and short-term monetary support, all expressed both in ECUs and in US dollars.

# Article 14 - Settlement of debts in reserve assets at the monthly settlement date (see Article 16 of the Agreement)

14.1 Method of repayment chosen by the debtor

Not later than five working days before the monthly settlement date, each central bank in debt to the Fund at that settlement date shall inform the Agent of the breakdown between ECUs and other reserve assets for the settlement that it proposes to make, subject to any advance repayments that might still be made under Article 10.2 of the present Directive.

14.2 Allocation of means of settlement to creditors

The Agent shall provisionally allocate to the creditor central banks the means of settlement furnished by the debtor central banks in accordance with the following principles:

- means of settlement furnished on "mutual agreement" accounts shall be used to repay claims on "mutual agreement" accounts with the same settlement date, and
- means of settlement furnished on "initial account" or "renewed debtor account" shall be used first to repay claims regarded as being of longest seniority on creditor accounts; with claims of equal seniority, these means of settlement shall be distributed among creditor central banks in proportion to the amount of their creditor balances.
14.3 Notification by the Agent of the method of repayment

14.3.1 The Agent shall notify each debtor central bank as soon as possible of:

- the amount in ECUs that will be debited to its "ECU reserves" account, and
- the breakdown of the other reserve assets the said debtor central bank will have to transfer for account of the Fund to each creditor central bank,

at the relevant monthly settlement date.

14.3.2 At the same time, the Agent shall inform each creditor central bank of:

- the amount in ECUs that will be credited to its "ECU reserves" account, and
- the composition of the other reserve assets that will be transferred to the said creditor central bank for account of the Fund by each debtor central bank,

at the relevant monthly settlement date.

14.3.3 The amounts of the reserve assets to be transferred shall be expressed in ECUs.

### Article 15 - Means of settlement (see Article 16 of the Agreement)

15.1 Transfer of ECUs

On the relevant date the Agent shall automatically effect the settlements in ECUs by transfer between the creditor and debtor central banks' "ECU reserves" accounts; at the same time, the Agent shall make the balancing entries on the ECU-denominated accounts in the Fund's books referred to in Article 8 above.

# 15.2 Transfer of assets denominated in SDRs

Debit balances to be settled by means of assets denominated in SDRs shall be converted into these assets on the basis of the rate for the ECU in US dollars, as referred to in Article 5.2 of the present Directive, and the rate for the US dollar in SDRs ruling on the third working day before the settlement day; since transfers of these reserve assets have to be effected by the IMF, it shall be the responsibility of the creditor and debtor central banks to make the necessary approaches to that institution with a view to having these transfers carried out in time and to notify the Agent of their execution.

## 15.3 Transfers of currency assets

In the case of settlement in currency assets, conversion of these assets into ECUs shall be effected at the rates ruling on the second working day before the settlement day.

#### 15.4 Balancing entries on accounts

It shall be the responsibility of the central banks with claims on or debts to the Fund to advise the Agent of the execution of settlements effected between them direct on the agreed date.

Upon receipt of the confirmations from the central banks concerned regarding the execution of the transfers of SDR-denominated assets and of the settlements in currencies effected directly between the central banks on the agreed date, the Agent shall make the balancing entries on the accounts opened in the Fund's books for the central banks concerned.

#### 15.5 Notification of balances

When the accounting entries referred to in Articles 15.1 to 15.4 above have been effected, the Agent shall notify the central banks concerned of the overall balance on their accounts in each relevant category, furnishing them also with a breakdown by maturity.

# III. <u>CREATION, UTILISATION AND REMUNERATION OF ECUs</u> (see Article 17 of the Agreement)

## Article 16 - Conclusion of swap operations

Acting in the name and for the account of the Fund, the Agent shall conclude with each of the central banks concerned the swap operations referred to in Article 2 of Decision No. 12/79, involving, on the spot side, the transfer of ECUs to the central banks against transfer by them to the Fund of the reserve assets defined in Article 17 below; these operations shall be in conformity with the standard contract contained in Annex I to the present Directive and shall be renewed for successive periods of three months on the terms and within the limits laid down in Articles 20 and 22 below.

# Article 17 - Definition of the reserve assets involved in the swap operations

#### 17.1 Percentage contributions

The weights of gold and the amounts of US dollars transferred in connection with the swap operations shall correspond to 20 per cent. of the gold assets and 20 per cent. of the gross US dollar assets of each central bank; these assets shall comprise

- the gold and US dollar reserves shown in the statement drawn up by each central bank as at the last working day of the month preceding the month in which the swap operations are concluded or renewed (see Article 13 of the present Directive) and
- the contributions to the Fund in accordance with Article 2.1 of Decision No. 12/79.

#### 17.2 Gold assets

The gold assets may include:

- a whole number of gold bars ranking as good delivery at the place of deposit; and
- holdings on gold account with a central bank or with the BIS.

#### 17.3 US dollar assets

The US dollar assets may include, at nominal value:

- balances on deposit with a central bank or with the BIS, or paper representing such deposits;
- Treasury bills and other securities issued by the US Treasury; and
- certificates or other securities issued by US Federal agencies.

## 17.4 Portion of 20 per cent. transferred to the Fund

The total of each of the categories of reserve assets transferred to the Fund shall be as near as possible to, but not less than, the 20 per cent. portion referred to under Article 17.1 above.

## 17.5 Communications concerning the reserve assets

17.5.1 The Agent shall receive communications from the central banks concerning the nature and identification of the reserve assets transferred by them to the Fund in connection with the swap operations.

17.5.2 The Agent shall keep up to date the statement of the reserve assets involved in the swap operations, the management of which is entrusted to the central banks (list of gold bars and statements of securities, etc.).

## Article 18 - Conversion rates to be applied

#### 18.1 ECU/gold conversion rate

The ECU/gold conversion rate, to be communicated by the Agent to the central banks on the first working day of each month, shall be calculated by the Agent, to three decimal places, on the basis of the average of the daily gold prices, converted into ECUs, recorded at the two London fixings during the six calendar months immediately preceding the month in which the swap operation was concluded or renewed; however, the average of the prices, converted into ECUs, at the two London fixings on the second last working day of the said period shall be used if it is lower than the abovementioned average of the prices recorded throughout that period.

## 18.2 US dollar/ECU conversion rate

The US dollar/ECU conversion rate shall be the rate established by the staff of the CEC two working days prior to the value date of the spot transaction concluded in connection with the swap operation (see Article 5.2 above).

## 18.3 Swap operations at flat rates

The same rates shall be applicable for the spot and the forward component of each swap operation.

## Article 19 - Management contracts

### 19.1 Standard contract

The management contracts concluded by the Agent with the central banks on behalf of the Fund in accordance with Article 1.2 above shall conform to the standard contract set out in Annex II to the present Directive.

## 19.2 Termination of contract

A management contract thus concluded shall expire upon maturity of the swap operations involving the assets which it covers.

# Article 20 - Quarterly renewals

20.1 Terms of renewal

Upon maturity, the swap operations and the corresponding management contracts shall be renewed on the same terms but at the rates applicable (see Article 18 of the present Directive) at the time of renewal and on the basis of a weight of gold and an amount of US dollars representing at least 20 per cent. of the corresponding reserves of the central bank concerned, including those reserve assets transferred to the Fund; the renewal shall be effected on the basis of the central bank's reserves as at the last working day of the preceding month.

20.2 Accounting operations on the occasion of renewal

On the occasion of renewal the Agent and the central banks shall effect the settlement of the differences in terms of ECUs, gold and US dollars resulting from the liquidation of the matured swap operations and the conclusion of the new swap operations.

20.3 Negative balance

If the Agent judges that the renewal of swap operations would produce a negative balance in ECUs, it shall immediately alert the central bank concerned in order that the latter may in good time take the necessary steps to prevent the emergence of such a balance.

# <u>Article 21</u> - Additional swap operations and termination in advance of maturity

## 21.1 Additional swap operations

At the request of a central bank giving two working days' notice, the Agent shall at any time conclude on behalf of the Fund additional swap operations for the amount specified by that central bank; the corresponding reserve assets shall represent equal proportions of the central bank's gold and US dollar assets as recorded in the statement of its reserves as at the last working day of the preceding month. These operations shall mature on the day of the following quarterly renewal and shall be renewable for successive periods of three months on the terms and within the limits laid down in Articles 20 and 22 of the present Directive; the other terms set out in Articles 16 to 19 of the present Directive shall apply to these operations.

# 21.2 Termination in advance of maturity

At the request of a central bank giving two working days' notice, the Agent shall at any time terminate part or all of the swap operations concluded with that central bank, on the sole condition that such termination does not cause the central bank to show a negative balance in ECUs; however, on the occasion of the next quarterly renewal fresh swap operations shall at all events be concluded in respect of at least 20 per cent. of the gold and US dollar reserves as laid down in Article 20 above.

#### Article 22 - Final termination

Save in the event of a unanimous decision to the contrary, the swaps of ECUs against gold and US dollars referred to in Article 17.3 of the Agreement shall be unwound at the end of the period for which the Governors of the central banks have extended the mechanism provided for in Article 17 of the said Agreement.

## Article 23 - Accounting procedures

#### 23.1 "ECU reserves" accounts

The Agent shall open for each central bank an "ECU reserves" account in the books of the Fund, which shall be credited with the ECUs issued by the Fund as the counterpart of the reserve assets involved in the swap operations; this account may not show a negative balance.

23.2 Transfers of ECUs

## 23.2.1 Matching instructions

Upon receipt of matching instructions from the central banks concerned indicating the amount and value date of the transfer, the Agent shall carry out transfers of ECUs in the books of the EMCF.

## 23.2.2 Intra-Community settlements

Within the framework of intra-Community settlements, the Agent shall also effect transfers of ECUs in the books of the EMCF in accordance with the provisions of Article 15 of the present Directive.

# 23.3 Payment of interest on net positions in ECUs

#### 23.3.1 Net positions

The Agent shall establish daily, for each central bank, its net position in ECUs, which shall correspond to the difference, either positive (net creditor position) or negative (net debtor position), between its spot assets on "ECU reserves" account and forward ECU claims on the EMCF, on the one hand, and its forward ECU liabilities vis-à-vis the EMCF on the other hand.

## 23.3.2 Calculation of interest

The interest on the daily net positions of each central bank shall be calculated by the Agent as at the last working day of each month on the basis of the rate determined in accordance with the procedure set out in Article 6 of the present Directive; the method to be applied shall be that of the exact number of days and 365 days a year.

23.3.3 Accounting procedures

On the last working day of each month the Agent shall make the accounting entries for purposes of settling the interest thus calculated by crediting and debiting the "ECU reserves" accounts of the central banks concerned; if it judges that this might produce a negative balance in ECUs, the Agent shall proceed in the manner laid down in Article 20.3 above.

#### IV. ECU MOBILISATION MECHANISM

## Article 24 - Mobilisable portion of allocated ECUs

At the time of the quarterly renewal of the ECU allocation operations in accordance with Article 17 of the Agreement the Agent shall calculate

- the mobilisable percentage (p) of allocated ECUs according to the following formula:

m being a variable multiplier set by the Governors of the central banks and notified to the Agent by the Secretariat of the Committee of Governors, and

- the theoretical mobilisable portion (TMP) of ECUs for each EMS central bank, defined as follows:

total volume of ECUs allocated to each central bank TMP = p x \_\_\_\_\_\_\_\_ 100

The Agent shall notify each central bank as soon as possible of the theoretical mobilisable portion of ECUs applicable in its case for the duration of the new ECU allocation operations. Article 25 - Procedure to be followed in the event of a mobilisation request by a central bank

#### 25.1 Notification of the mobilisation operation

At the latest by 2 p.m. (Brussels time) on the fourth working day before the value date of the mobilisation operation, the mobilising central bank shall notify the Agent of its wish to mobilise against US dollars part of its assets in ECUs, indicating

- the amount of ECUs to be mobilised,
- the value date of the transaction,
- where applicable, the portion of the proceeds of the mobilisation operation in question which it wishes to convert into one or more EMS currencies,
- the account or accounts to which the currency transfers are to be made.

## 25.2 Steps to be taken by the Agent

25.2.1 Upon receipt of such notification the Agent shall establish that the amount of ECUs to be mobilised is not greater than

- the actual mobilisable portion (AMP) of ECUs available to the central bank in question, calculated according to the following formula for the value date concerned:

	+ net creditor position	ECUs already mobilised
AMP = TMP	or	by that central bank
	- net debtor position	- )

- the financing resources available, i.e. the ECU equivalent of the US dollars transferred to the EMCF by all the EMS central banks with the exception of the mobilising central bank in accordance with Article 2 of Decision No. 12/79, minus the ECU equivalent of the US dollars already transferred in connection with outstanding mobilisation transactions.

25.2.2 Once it has established this, the Agent shall advise the mobilising central bank that its request conforms to the rules and that it (the Agent) will take the necessary steps to cover the request as far as possible.

25.2.3 The Agent shall calculate the ECU amounts of the other EMS central banks' individual cover contributions (CCs) in accordance with the following formula:

commitment of the contributing central bank to provide US dollars to the EMCF

CC = amount of ECUs mobilised x

total commitment of all EMS central banks with the exception of the mobilising central bank to provide US dollars to the EMCF

The commitment represents the net amount of US dollars transferred to the EMCF in accordance with Article 2 of Decision No. 12/79 (i.e. minus any amount which may have been withdrawn in accordance with Article 3.3 of that Decision), minus any US dollars already provided in connection with outstanding mobilisation transactions.

25.2.4 The Agent shall

- inform the other EMS central banks of the request for mobilisation and, if such is the case, for conversion into one or more EMS currencies, indicating their cover contributions (CCs) in ECUs,
- ask them to send it, not later than midday (Brussels time) three working days prior to the value date of the operation, confirmation of their participation in the mobilisation operation;

in the event of conversion into one or more EMS currencies, it shall also

- ask each central bank issuing a desired EMS currency
  - . whether it has any objection to conversion into its currency, and,
  - . if it has none, the amount, expressed in ECUs, up to which it is willing to furnish its currency in connection with the conversion operation, and
- ask the other EMS central banks the amounts, expressed in ECUs, in the EMS currencies up to which they would be willing to participate in the conversion operation in the event that the central bank issuing the desired currency has no objection to the conversion.

25.2.5 As early as possible on the afternoon of the fourth working day prior to the value date of the mobilisation operation the Agent shall take all the steps provided for under Articles 25.2.1 to 25.2.4 above.

## 25.3 Opting-out of cover operations

If a contributing central bank decides to opt out of the cover operations, entirely or in part, it shall inform the Agent of its decision within the same deadline as that set in Article 25.2.4 above; in this event the Agent shall reallocate the cover contributions, notifying the contributing central banks, not later than 2 p.m. (Brussels time) on the third working day prior to the value date of the operation, of the new amounts in ECUs of their contributions and requesting them to confirm their agreement that same afternoon.

#### 25.4 Insufficient cover

To the extent that the mobilisation request cannot be met out of the commitments of the contributing central banks, it shall be automatically scaled down by the necessary amount by the Agent, which shall advise the mobilising central bank accordingly before the close of business on the third working day prior to the value date of the operation.

# 25.5 Notification by the Agent of the definitive amounts of the cover contributions in ECUs

Before the close of business on the third working day prior to the value date of the mobilisation operation the Agent shall advise the participating central banks of the definitive amounts in ECUs of their contributions to covering the said operation.

# Article 26 - Conversion into EMS currencies

If the mobilising central bank has requested the immediate conversion of all or part of the proceeds of the mobilisation operation, and each central bank issuing a desired currency has given the Agent confirmation of its consent to such conversion (see Article 25.2.4 above), the Agent shall allocate the EMS currency amounts among the participating central banks as follows:

(a) if the total of the amounts offered for conversion is greater than the amount requested by the mobilising central bank, the amount offered by the central bank issuing the currency into which the conversion is to be made shall be reduced accordingly up to the point where it may be left out of account completely; (b) if there is still a balance available, the offers received from the other contributing central banks shall be reduced in proportion to the size of their offered contributions;

(c) on the other hand, if the amount requested can only be met in part, it shall be reduced automatically by the Agent, which shall inform the mobilising central bank of this reduction before the close of business on the third working day prior to the value date of the operation.

# Article 27 - Conclusion of mobilisation and conversion operations

## 27.1 ECU/US dollar (mobilisation) swaps

As soon as the exchange rates referred to in Article 5.2 above have been reported to it, in the course of the afternoon of the day mentioned under Article 27.5 below, the Agent, acting in the name and for the account of the Fund, shall conclude a swap operation with the mobilising central bank, selling spot US dollars against ECUs, and cover this transaction through simultaneous swap operations with the contributing central banks, buying spot US dollars against ECUs.

# 27.2 US dollar/EMS currency (conversion) swaps

To the extent that the US dollar proceeds are to be converted into one or more EMS currencies, the Agent shall conclude a swap of US dollars against the EMS currency or currencies with the mobilising central bank, buying spot US dollars from it, and cover this transaction by selling these US dollars to the contributing central banks against the EMS currency or currencies required under this swap operation.

### 27.3 Conversion rates

All the swap operations shall be concluded at flat rates, applying

- for the mobilisation swaps, the US dollar/ECU rate ruling two working days before the value date of the spot operation (see Article 5.2 of the present Directive), and
- for the conversion swaps, the US dollar/EMS currency conversion rates obtained by calculating the cross rate of the above US dollar/ECU rate against the ECU rate for the currency into which conversion is to be made, as notified by the CEC the same day.

#### 27.4 <u>Term</u>

Each mobilisation or conversion swap operation shall be concluded for a term of three months, irrespective of the date of the quarterly renewal of the swaps concluded in accordance with Article 17 of the Agreement.

27.5 Notification of the details of the swap operations

As early as possible on the afternoon of the second working day prior to the value date of the mobilisation operation, the Agent shall advise the participating central banks of the details of these swap operations, asking them to send it confirmation of their agreement.

#### Article 28 - Transfers

## 28.1 Transfer of currency assets

The transfers of currency assets (US dollars and EMS currencies) necessitated by the mobilisation and conversion swaps with the Fund shall be executed direct by the participating central banks among themselves; the central banks concerned shall

- furnish each other with the details necessary for executing the transfers,
- send the Agent confirmation of receipt of the currencies in question.

28.2 Transfer of ECUs

The transfers of ECUs necessitated by the mobilisation operations shall be effected by the Agent by debiting and crediting the "ECU reserves" accounts of the central banks concerned once it has received confirmation of the receipt of the currency funds by the respective central banks.

Article 29 - Remuneration on mobilisation and conversion operations

### 29.1 Interest rates

Interest shall be paid on the forward currency liabilities and claims arising from the swap operations referred to in Article 27 above at the rates referred to in Article 6.4 above, which shall remain unchanged for the duration of the swap operations.

# 29.2 Calculation and notification by the Agent

29.2.1 The interest on the forward currency liabilities and claims of each central bank shall be calculated by the Agent at the maturity of the swap operations in question or upon their termination in advance of maturity; the method to be applied shall be that of the exact number of days and 365 days a year. The Agent shall notify the central banks concerned of the amounts thus calculated.

29.2.2 The interest shall be added to the amounts transferred in respect of principal at the maturity of the swap operations in question or upon their termination in advance of maturity and shall be settled on the same terms.

Article 30 - Renewal of mobilisation and conversion operations

30.1 Conditions of renewal

At the request of the mobilising central bank the swap operations referred to in Article 27 of the present Directive may be renewed once for three months; in this event the Agent, together with the central banks participating in these new operations, shall renew the mobilisation and conversion operations, it being understood that:

- the new operations shall be concluded at the exchange rates and interest rates applying at the time of renewal (see Articles 27.3 and 29.1 of the present Directive);
- the amount of ECUs to be mobilised shall not exceed either the mobilising central bank's actual mobilisable portion (AMP) on the value date of the new swap operation or the ECU equivalent of the US dollar contributions made by the central banks under Article 3a.2 of Decision No. 12/79; if the amount in ECUs of the matured operation exceeds these limits, it shall automatically be scaled down accordingly;
- the contributions of US dollars made by the central banks under Article 3a.2 of Decision No. 12/79 shall be adjusted both in proportion to and within the limits of the commitments defined there and to take account of exchange rate changes that have occurred since the initial conclusion of the maturing mobilisation swap; and
- the amounts of EMS currencies shall be adjusted to take account of exchange rate changes that have occurred since the initial conclusion of the maturing conversion swap.

## 30.2 Application of the provisions of Article 25

The provisions of Article 25 of the present Directive shall apply by analogy to the renewal operations.

#### 30.3 Settlement of differences

At the time of these renewals, the Agent and the central banks concerned shall settle the differences resulting from the liquidation of the matured swaps and the conclusion of the new swaps.

## Article 31 - Mobilisation of the balance of ECUs

A central bank which has mobilised ECUs up to its ceiling shall still be entitled to utilise its balance of ECUs on "ECU reserves" account for settlements or in voluntary transactions without being obliged to unwind the mobilisation operation before maturity; similarly, it shall still be entitled to unwind a swap operation in accordance with Article 3.3 of Decision No. 12/79.

Article 32 - Termination of mobilisation and conversion swap operations

## 32.1 Termination in advance of maturity

32.1.1 With at least three working days' notice given by the mobilising central bank, all or part of mobilisation and conversion operations may be terminated in advance of the initial settlement or renewal date.

32.1.2 Upon receipt of such notice the Agent shall inform the central banks participating in the operation of its termination in advance of maturity, giving them the details of the unwinding of the swap operation and the payment of interest due.

## 32.2 Termination arrangements

At the initial or extended settlement date of mobilisation and conversion operations, or upon their termination in advance of maturity, the Agent and the central banks shall liquidate these operations in accordance with the terms and conditions set out in Articles 27 and 28 of the present Directive.

## V. OTHER HOLDERS OF ECUS CREATED BY THE EMCF

Article 33 - Opening of accounts in the name of Other holders of ECUs

The Agent shall open in the books of the EMCF an "ECU reserves" account in the name of each Other holder of ECUs of which it is notified by the Secretariat of the Committee of Governors.

## Article 34 - Terms and conditions governing transactions in and holdings of ECUs by Other holders of ECUs

# 34.1 ECU accounting procedures and notification of account balances by the Agent

Transfers of ECUs between a central bank and an Other holder of ECUs, or between two Other holders of ECUs, shall be entered in the books of the EMCF by the Agent on receipt of matching instructions from the institutions concerned, by analogy with Article 23.2.1 of the present Directive.

# 34.2 Payment of interest on holdings of ECUs

The Agent shall

- calculate, as at the last working day of each month, the accrued interest on the "ECU reserves" account of the Other holder of ECUs, notifying the holder accordingly, and
- credit this interest to the said account on the last working day of each month.

The provisions of Article 23.3 of the present Directive shall apply by analogy.

Article 35 - Closing of accounts held in the name of Other holders of ECUs

35.1 <u>Closing of accounts on suspension of the status of</u> Other holder of ECUs

The Secretariat of the Committee of Governors shall inform the Agent of any suspension of the status of Other holder of ECUs; in such a case the account of the Other holder of ECUs concerned shall be closed as soon as possible.

35.2 <u>Closing of accounts in the event of the winding-up of the ECU</u> creation mechanism

All accounts of Other holders of ECUs shall be closed upon the winding-up of the ECU creation mechanism.

Annex I

#### SWAP OPERATIONS

Telex from the central bank to the Agent on the occasion of the quarterly renewal of swap operations

FECOM No.

1. Termination of swap operations maturing on .....

We confirm our agreement with the arrangements made in accordance with PRIMO of your FECOM No. ... of .....

#### PRIMO

Our reserves as at ..... (including contributions of reserves to the EMCF)

#### SECUNDO

(a) All value .....

We sell to the EMCF ...... ounces of fine gold in bars, good delivery at the place of deposit, against ECU ..... at a price of ECU ..... per fine ounce and request that you credit the amount in ECUs to our "ECU reserves" account in the books of the EMCF.

[The gold bars comprise those already mentioned in the lists/letters/telexes previously sent by us to the EMCF and ..... additional bars, specifications of which will be given to you as soon as possible.]

These bars will be placed at the disposal of the EMCF and held in accordance with its instructions.

(b) All value .....

We buy from the EMCF the same total weight of gold in fine ounces against ECU ..... at a price of ECU ..... per fine ounce and request that you debit our "ECU reserves" account in the books of the EMCF with the amount in ECUs. Please place the corresponding gold bars at our disposal.

(c) This swap operation may at any time be wholly or partially unwound at our request, subject to two working days' notice.

#### TERTIO

(a) All value .....

We sell to the EMCF US\$ ..... against ECU ..... at a price of 1 ECU = US\$ ..... and request that you credit the amount in ECUs to our "ECU reserves" account in the books of the EMCF.

[The US dollar assets in question comprise those sold to the EMCF under the previous swap operation and managed by us for its account, with the exception of those for which a detailed statement will be sent to the EMCF as soon as possible.]

The US dollar assets will be placed at the disposal of the EMCF and held in accordance with its instructions.

(b) All value .....

We buy from the EMCF US\$ ..... against ECU ..... at a price of 1 ECU = US\$ ..... and request that you debit our "ECU reserves" account in the books of the EMCF with the amount in ECUs.

Please place the corresponding US dollar assets at our disposal.

(c) This swap operation may at any time be wholly or partially unwound at our request, subject to two working days' notice.

#### QUARTO

Please confirm agreement.

Annex II

1

#### MANAGEMENT CONTRACT

Telex from the Agent to the central bank after the conclusion of the swap operations

FECOM No.

Article 19 of the Sixth Directive

#### PRIMO

We confirm our agreement with your telex FECOM No. ..... of .....

#### SECUNDO

Please manage on behalf of the EMCF the gold and US dollar assets sold to the EMCF pursuant to the swaps referred to in your telex under primo until those swaps mature.

#### TERTIO

You are hereby authorised to change the composition of the assets with whose management you are entrusted in accordance with secundo above and which are set forth under secundo (a) and tertio (a) of your telex mentioned under primo, provided that

- the total weight of gold and the nominal value of the US dollar assets are not reduced by any such change,
- the composition of the total assets in question remains in conformity with the terms of Article 17 of the Sixth Directive,
- the effects of the aforementioned swaps automatically extend to any such replacement assets.

Please advise us by telex as soon as possible of any such changes made in accordance with these conditions.

#### QUARTO

Whenever appropriate, it shall be for you to notify third-party holders of the assets entrusted to your management of their transfer to the EMCF, and any such notification shall be effected by you on your sole responsibility and without our having to take any action in the matter.

#### QUINTO

The income arising from the assets under your management shall be collected by and accrue to you. The present contract shall give rise to no other remuneration. The costs of managing the aforesaid assets, and likewise any taxes, duties or other charges relating thereto, shall be borne by you.

#### SEXTO

Please confirm agreement.

EMCF Agent

Committee of Governors of the Central Banks of the Member States of the European Economic Community

\_

· ``

Secretariat

\_

# ECU REMUNERATION

# 1. LIST OF REFERENCE RATES

BELGIUM:	Three-month interbank deposits rate
DENMARK:	Three-month money market rate
GERMANY:	Three-month interbank deposits rate
GREECE :	Rate for three-month interbank deposits in convertible drachmae
FRANCE :	Three-month interbank money rate against private paper
IRELAND:	Three-month interbank rate
ITALY:	Tender rates for three-month Treasury bills
LUXEMBOURG:	see Belgium
NETHERLANDS:	Three-month interbank deposits rate
UNITED KINGDOM:	Market yields for three-month UK Treasury bills

# 2. CALCULATION OF THE MONTHLY AVERAGE OF DAILY OBSERVATIONS

Belgium, Germany, Greece, Ireland, the Netherlands and the United Kingdom calculate an unweighted arithmetic average over the number of working days in the month.

Denmark and France calculate an unweighted average on the basis of all calendar days, including weekends and holidays, for which the rate on the preceding working day is used.

Italy calculates an average of tender rates (if there is more than one during the month concerned) which is weighted on the basis of the respective sales of Treasury bills.

- 2 -

### 3. EXAMPLE ILLUSTRATING THE EFFECTS OF THE NEW FORMULA FOR ECU REMUNERATION \_\_\_\_\_

Country	Currency weights (1)	Representative domestic money market rates (2)	Weighted average
Belgium Luxembourg	)  ) 8.57446  )	10.7888	0.92508
   Denmark	2.69007	11.97	0.32200
Germany	32.07187	6.39	2.04939
Greece	   1.31457	27.45	0.36085
France	19.05577	10.73	2.04468
Ireland	1.20024	14.39	0.17271
Italy	   9.97513	12.90	1.28679
Netherlands	10.13480	7.19	0.72869
United Kingdom	   14.98314 	13.38	2.00474
Total	100.00005	-	9.89493

Interest rate which would have been applicable in April 1985: 9.75% p.a. (3)

Actual interest rate applicable in April 1985: 8.75% p.a. (3)

(1) Derived from ECU central rates applicable since 17th September 1984. • (2) Monthly average of daily observations in March 1985.

(3) Weighted average rounded down to the nearest 1/4 per cent.

- 3 -

÷.,

Committee of Governors of the Central Banks of the Member States of the European Economic Community

## GROUP OF EXPERTS UNDER THE CHAIRMANSHIP OF MR. DALGAARD

### List of Participants

Chairman

Banque Nationale de Belgique

Danmarks Nationalbank

Deutsche Bundesbank

Bank of Greece

Banque de France

Central Bank of Ireland

Banca d'Italia

De Nederlandsche Bank

Bank of England

Institut monétaire luxembourgeois Commission of the European Communities

Secretariat of the Monetary Committee Agent for the EMCF

Secretariat of the Committee of Governors

H. Dalgaard P. Adriaens

J. Michielsen

- H. Fosgaard
- 0. Hollensen
- A. Kloft K. Weber

P. Pliatsikas

- A. Icard
- B. Marillia
- G. McGrath

P. Charleton

- F. Saccomanni
- S. Rebecchini

J.W. Brockmeijer A.F.P. Bakker D.A.H. Byatt J.E.W. Kirby

J. Guill

A. Louw

H. Lohan

R. Barthélémy

J.C. Dagassan K. Flury

A. Bascoul H.K. Scheller A.W. Mortby

The group held one meeting on 15th and 16th April 1985.